. . CSC 560 Modern DBMS Architectures Alexander Dekhtyar . . Assignment 4: Rock-Paper-Scissors-Lizard-Spock

Due date: Tuesday, November 6

Assignment

Each team will do this:

1. Upload the final version of their project proposal to the wiki.

2. Study the ”Rock-paper-scissors-lizard-Spock” diagram below, determine which other teams it provides the comments for and which team will provide the comments for it.

3. By Thursday, November 1, read the two proposals of the teams it needs to proivde the comments to, and prepare and upload the list of comments.

4. By Tuesday, November 6: receive comments from two other teams, discuss them and prepare a short response. Upload the response on the wiki.

Good Luck!